**Author's response to reviews**

**Title:** Relationship between working with Video Display Terminals (VDT) and frequency of dermatologic manifestations in office workers: a cross-sectional study

**Authors:**

Omid Aminian (oaminian@sina.tums.ac.ir)
Parvin Mansoori (ilcpm@hotmail.com)
Akbar Sharifian (k.sharifian@yahoo.com)
Ehsan Rafeemanesh (rafeemanesh@razi.tums.ac.ir)
Mohamadreza Iraniha (iraniham@sina.tums.ac.ir)

**Version:** 7  **Date:** 7 December 2004

**Author's response to reviews:**

Dear Reviewer

As diagrams 1 & 2 and tables 1 & 2 showed the same figures therefore diagrams were omitted.

With statistical analysis the confounding factors (age and sex) were compared between the two groups and no statistically significant difference was observed.

Mean exposure time was compared between the two groups, using statistically analysis and there was no statistically significant differences.

In table(3) we used the chi-square, linear by linear association test instead of cochrane armitage test and found statistically significant relationship between increasing weekly exposure time and dermatologic manifestations. (P.value < 0.05)

Miss spellings were corrected in the tables.

The references list was corrected as much as possible.

The hole manuscript has been edited completely.

Dear